## TROPICAL AGRICULTURAL RESEARCH Author Response to Reviewer Comments

Reviewer comment/Suggested amendment	Attended, if so, line & Page Numbers	Not attended, if so, justification for not attending
Reviewer 1		
1		
2		
3		
4		
5		
6		
(Page add additional responses if required)		
Reviewer 2		
1		
2		
3		
4		
5		
6		
(Page add additional responses if required)		
Reviewer 3		
1		
2		
3		
4		
5		
6		
(Page add additional responses if required)		

Name of	Corresponding Author	r ·

Signature -Date -